Title: Potential benefits of using a toolkit developed to aid in the adaptation of HTA reports: a case study considering positron emission tomography (PET) and Hodgkin's disease

Authors:

Sheila Turner (s.turner@soton.ac.uk)
Neil Adams (neil.adams@hpa.org.uk)
Andrew Cook (andrewc@soton.ac.uk)
Alison Price (alison@soton.ac.uk)
Ruairidh Milne (rm2@soton.ac.uk)

Version: 4 Date: 12 May 2010

Author's response to reviews:

Covering letter to the Editors

Revisions made 11th May 2010:

Authors: name changed to read Sheila Turner

Tables:

• Vertical text elements changed to horizontal.
• Shading removed
• Vertical lines removed

Title: changed to sentence case.

Title page: submission information deleted

Placement text: removed

Abbreviations: formatted into a paragraph

Typography: Manuscript checked for typography errors. None identified.

Many thanks

Sheila Turner